Author’s response to reviews

Title: Open reduction and internal fixation of humeral midshaft fractures: anterior versus posterior plate fixation

Authors:

Sebastian Lotzien (sebastianlotzien@hotmail.com;sebastian.lotzien@bergmannsheil.de)
Clemens Hoberg (Clemens.Hoberg@bergmannsheil.de)
Valentin Rausch (Valentin.rausch@bergmannsheil.de)
Thomas Rosteius (Thomas.Rosteius@bergmannsheil.de)
Thomas Schildhauer (Thomas.schildhauer@bergmannsheil.de)
Jan Gessmann (Jan.Gessmann@bergmannsheil.de)

Version: 3 Date: 09 Oct 2019

Author’s response to reviews:

Manuscript number: BMSD-D-19-00728

Open reduction and internal fixation of humeral midshaft fractures: anterior versus posterior plate fixation
Sebastian Lotzien, MD; Clemens Hoberg, MD; Valentin Rausch, MD; Thomas Rosteius, MD; Thomas Armin Schildhauer, PhD; Jan Gessmann, PhD

Dear reviewers, dear assistant editor

Thank you very much for the review of the manuscript.

Please find our answers to your comments and questions below.

Reviewer reports:

Assistant Editor Comments:

1. Funding
In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

We revised the funding section of the manuscript:
Lines 385-389: The publication charges for this article have been funded by the DFG Open Access
Publication Funds of the Ruhr-Universität Bochum. The funding body was neither involved in the design of the study nor in collection, analysis, and interpretation of data and in writing the manuscript.